Author’s response to reviews

Title: Knowledge about iodine requirements during pregnancy and breastfeeding among pregnant women living in Northern Ireland

Authors:

Paul McMullan (paul.mcmullan@southerntrust.hscni.net)
Alyson Hunter (alyson.hunter@belfasttrust.hscni.net)
David McCance (david.mccance@belfasttrust.hscni.net)
Jayne Woodside (j.woodside@qub.ac.uk)
Karen R Mullan (karen.mullan@belfasttrust.hscni.net)

Version: 3 Date: 09 Feb 2019

Author’s response to reviews:

Editor Comments:

1. While your ethics committee was clarified to be the Office of Research Ethics Committee Northern Ireland in the Methods, this is not clarified in the Ethics approval and consent to participate subsection of the Declarations. Please revise accordingly.
Completed

2. Please also take this opportunity to add the detail to the Ethics approval and consent to participate subsection of the Declarations that consent was provided verbally and that this method was approved by your ethics committee.
Completed

3. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.
Completed
4. The individual names and contributions of authors to the manuscript should be specified in the Authors’ contributions section of the Declarations. Please note that all authors should be named by using their initials rather than full names, and also include a statement clearly indicating that all authors read and approved the final version of the manuscript.

Completed

5. Please add a section "Consent for publication" in the Declarations. You may state "Not applicable" in this section as no personal information is provided in your manuscript.

Completed

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Completed